Author's response to reviews

Title: Psychological illness is commonly associated with functional gastrointestinal disorders and is important to consider during patient consultation: a population-based study

Authors:

Ture Alander (ture@tapraktik.se)
Kurt Svardsudd (kurt.svardsudd@pubcare.uu.se)
Sven-Erik Johansson (imse.johansson@swipnet.se)
Lars Agreus (lars.agreus@klinvet.ki.se)

Version: 7 Date: 27 April 2005

Author's response to reviews:

COVER LETTER
The references have been completed with all author names and formatted according to Biomed Central author instructions. A sentence on page 13 has been divided into two for a clearer understanding.